Author's response to reviews

Title: A Danish population-based cohort study of newly diagnosed asthmatic children's care pathway - adherence to guidelines

Authors:

Grete Moth (grete.moth@alm.au.dk)
Peter O Schiotz (poa@sks.aaa.dk)
Peter Vedsted (p.vedsted@alm.au.dk)

Version: 5  Date: 10 June 2008

Author's response to reviews:

Dear editor

All the required formatting changes have been made and are listed below:

Author list - updated
Country - added
Key words - deleted
Abbreviations - renamed and moved
Box 1 - the text in box 1 has been written into the text
Author's contributions - revised
Appendices - moved and tables changed into text
Table orientation - the big Table 2 has been changed into 3 tables: Tables 2-4

I hope this will suffice

Yours sincerely

Grete Moth